Author’s response to reviews

Title: Mother-Infant Bonding is not Associated with Feeding Type: A Community Study Sample

Authors:

Ilana Hairston (hanahai@telhai.ac.il)
Jonathan Handelzalts (jonathanh@013.net)
Tamar Lehmen-Inbar (tamar.lehman@gmail.com)
Michal Kovo (michalkovo@gmail.com)

Version: 5 Date: 23 Mar 2019

Author’s response to reviews:

We have made the following corrections or verifications per your request:

1. All table titles should be in one line table number, and abbreviations are include in the table footnote. Please note that for tables NOT inserted in the body of the text, the notes are in the 'Legends' section.

2. Conclusions heading was added, and the text adjusted appropriately.

3. The role of the funding body was added.

4. A statement clearly indicating that all authors read and approved the final version of the manuscript, was added.

5. A section “Additional Files” after References was added

6. The manuscript was proofread again, and minor typos and punctuation were corrected.